Author’s response to reviews

Title: The impact of Immigration Detention on Mental Health: A Systematic Review

Authors:

Martha von Werthern (martha.vonwerthern@helenbamber.org)
Katy Robjant (katy.robjant@vivo.org)
Zoe Chui (zoeemilychui@googlemail.com)
Rachel Schon (rachel.h.schon@gmail.com)
Livia Ottisova (livia.ottisova@kcl.ac.uk)
Claire Mason (claire.mason@helenbamber.org)
Cornelius Katona (cornelius@helenbamber.org)

Version: 2 Date: 08 Aug 2018

Author’s response to reviews:

Thanks

We have provided a new 'response to reviewers' document in which each response is addressed separately and line references are given for the changes

We have also taken the opportunity to correct a couple of small errors in the manuscript.